Author's response to reviews

Title: Cesarean and VBAC rates among immigrant vs. native-born women: a retrospective observational study from Taiwan

Authors:

Jung-Chung Fu (ufifuh@yahoo.com.tw)
Sudha Xirasagar (sxirasagar@sc.edu)
Jihong Liu (jliu@mailbox.sc.edu)
Janice C Probst (jprobst@sc.edu)

Version: 10 Date: 9 September 2010

Author's response to reviews: see over
Dear Editor:
We have attended to all requested items.

1. All references are now formatted to the style of BMC Public Health illustrated in your sample manuscript
2. Reference #11 is now provided with an active webpage url.
3. The title sentence case is now corrected as per your sample manuscript format.

I hope this satisfies the requirements of BMC Public Health. Please do not hesitate to let us know if we need to do anything else. Thank you again for the opportunity to share our work with your readers.

Best regards,
Sudha Xirasagar